Author's response to reviews

Title: Torsades de pointes during laparoscopic adrenalectomy of a pheochromocytoma: a case report

Authors:

Kinge K van der Heide (k.van.der.heide@cso.umcg.nl)
Ann A de Haes (anndehaes@gmail.com)
Götz JKG Wietasch (J.K.G.Wietasch@anest.umcg.nl)
Ans ACP Wiesfeld (A.C.P.Wiesfeld@thorax.umcg.nl)
Herman HGD Hendriks (H.G.D.Hendriks@anest.umcg.nl)

Version: 4 Date: 25 March 2011

Author's response to reviews: see over
- Abstract restructured into 'introduction/case presentation/conclusion. Also confirmed it to the journal style, guided by the 'instructions for authors'

- Ethnicity of the patient included in the case presentation section.

- Figure 1c is adjusted